Author’s response to reviews

Title: Ethical issues in implementation research: a discussion of the problems in achieving informed consent

Authors:

Jane L Hutton (J.L.Hutton@warwick.ac.uk)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)

Version: 8 Date: 16 December 2008

Author’s response to reviews:

I have re-ordered the references in the text; added references to the title lines of the three descriptions of the trials (only one of these was done previously); re-ordered the references in the reference list; removed the only occurrence of et al that I could find (reference 2); ensured that cited references in the text and list match.

I also took the opportunity to replace former reference 12 with a new reference (now 14) which is the published paper for that study rather than the trial report.